%\documentstyle[a4,twocolumn]{article}
\documentstyle[a4]{article}

\def \mathfbox#1{\begin{array}{|c|}
			\hline \\
			\displaystyle  #1   \\[2ex]
   			\hline 
			\end{array}}

 
\title{Quantum kicked rotors and related systems}
\author{Hiroshi FUJISAKI \thanks
{Institute of Physics, University of Tokyo,
%3-8-1 Komaba, Tokyo 153, Japan, 
e-mail: fujisaki@ASone.c.u-tokyo.ac.jp}}
%\date{}
\begin{document}
\begin{thebibliography}{99}

\bibitem{Casa}
G.\ Casati, B.\ V.\ Chirikov, F.\ M.\ Izraelev, and J.\ Ford:
``Stochastic behavior of a quantum pendulum under a periodic perturbation", 
in {\it Stochastic Behavior in Classical and Quantum Hamiltonian Systems},
ed. by G.\ Casati and J.\ Ford, Lecture Notes in Physics, Vol. 93, Springer, Berlin (1979).

\bibitem{Zasl}
G.\ M.\ Zaslavsky: ``Stochasticity in Quantum Systems'',
Phys.\ Rep.\ {\bf 80}, 157-250 (1981);
{\it Chaos in Dynamic Systems}, Harwood Academic,
New York (1984).

\bibitem{Ott}
E.\ Ott: {\it Chaos in Dynamical Systems}, 
Cambridge University Press, Cambridge (1993).

\bibitem{Chir}
B.\ V.\ Chirikov: ``A Universal Instability of Many-dimensional 
Oscillator Systems", Phys.\ Rep.\ {\bf 52}, 263-379 (1979).


\bibitem{IS}
F.\ M.\ Izraelev and D.\ L.\ Shepelyansky: 
``Quantum Resonance for a Rotator in a Nonlinear Periodic Field",
Sov.\ Phys.\ Dokl.\ {\bf 24}, 996- (1979).

\bibitem{QS}
K.\ S.\ J. Nordholm and S.\ A.\ Rice:
J.\ Chem.\ Phys.\ {\bf 61}, 203-, 768- (1974);
K.\ Hepp and E.\ H.\ Lieb: in Lecture Notes in Physics, Vol. 38,
ed. by J.\ Moser, Springer, New York (1974);
A.\ Connes and E.\ Stormer: Acta Mathematica {\bf 134}, 289- (1975).


\bibitem{Fish}
S.\ Fishman, D.\ R.\ Grempel, and R.\ E.\ Prange:
``Chaos, Quantum Recurrences, and Anderson Localization",
Phys.\ Rev.\ Lett.\ {\bf 49}, 509 (1982);
D.\ R.\ Grempel, R.\ E.\ Prange, and S.\ Fishman:
``Quantum dynamics of a nonintegrable system",
Phys.\ Rev.\ A {\bf 29}, 1639-1647 (1984).

\bibitem{Ishi}
K.\ Ishii: ``Localization of Eigenstates and Transport Phenomena
in the One-Dimensional Disordered System'',
Prog.\ Theor.\ Phys.\ Suppl.\ {\bf 53}, 77-138 (1973).

\bibitem{Lee}
P.\ A.\ Lee and T.\ V.\ Ramakrishnan:
``Disordered electronic systems'',
Rev.\ Mod.\ Phys.\ {\bf 57}, 287-337 (1985);
B.\ Kramer and A.\ MacKinnon: 
``Localization --- Theory and Experiment",
Rep.\ Prog.\ Phys.\ {\bf 56}, 1469-1564 (1993).

\bibitem{Kitt}
C.\ Kittel: {\it Introduction to Solid-State Physics}, 
6th edition, John Wiley \& Sons (1986).

\bibitem{Chai}
G.\ J.\ Chaitin: ``Randomness and Mathematical Proof",
Scientific American, June p.47-52 (1975).

\bibitem{Penr}
R.\ Penrose: {\it The Emperor's New Mind
--- Concerning computers, minds, and the laws of physics}, Penguin (1991).

\bibitem{Gell}
M.\ Gell-Mann: 
{\it The Quark and the Jaguar --- Adventures in the simple and the complex}, 
Abacus (1995).

\bibitem{AZ}
A.\ Altland and M.\ R.\ Zirnbauer: 
``Field theory of the quantum kicked rotor", 
%preprint chao-dyn/9609014;
Phys.\ Rev.\ Lett.\ {\bf 77}, 4536 (1996).

\bibitem{Efet}
K.\ B.\ Efetov:
``Supersymmetry and theory of disordered metals", 
Adv.\ Phys.\ {\bf 32}, 53-127 (1983);
{\it Supersymmetry in Disorder and Chaos}, 
Cambridge University Press, Cambridge (1997).

\bibitem{TAI}
M.\ Toda, S.\ Adachi and K.\ Ikeda:
``Dynamical Aspects of Quantum-Classical Correspondence in Quantum Chaos",
Prog.\ Theor.\ Phys.\ Suppl.\ {\bf 98}, 323-375 (1989).

\bibitem{Four}
E.\ Abraham, P.\ W.\ Anderson, D.\ C.\ Licciardello, and T.\ V.\ Ramakrishnan:
``Scaling Theory of Localization: Absence of Quantum Diffusion 
in Two Dimensions'',
Phys.\ Rev.\ Lett.\ {\bf 42}, 673 (1979).

\bibitem{Shep}
D.\ L.\ Shepelyansky: 
``Localization of Quasienergy Eigenfunction in Action Space",
Phys.\ Rev.\ Lett.\ {\bf 56}, 677 (1986).

%\bibitem{Bran}
%T.\ Brandes: private communication.

\bibitem{CGS}
G.\ Casati, I.\ Guarneri, and D.\ L.\ Shepelyansky: 
``Anderson Transition in a One-Dimensional System 
with Three Incommensurate Frequencies",
Phys.\ Rev.\ Lett.\ {\bf 62}, 345 (1989).

\bibitem{Shep2}
D.\ L.\ Shepelyansky:
%``Localization of diffusive excitation in multi-level systems'', 
%Physica D {\bf 28}, 103 (1987);
Physica D {\bf 8}, 208 (1983);
``Coherent Propagation of Two Interacting Particles in a Random Potential",
Phys.\ Rev.\ Lett.\ {\bf 73}, 2607 (1994).

\bibitem{Iked}
K.\ Ikeda: ``Self-Organization of Dissipation in 
Classically Chaotic Quantum Systems", 
Ann.\ Phys.\ (N.Y.) {\bf 227}, 1-75 (1993).

\bibitem{Gasp}
P.\ Gaspard: ``Dynamical chaos and many-body quantum systems",
in {\it Quantum Chaos --- Quantum Measurement}, ed. by 
P.\ Cvitanovi\'c, I.\ Percival, and A.\ Wirzba, p.19-42, Kluwer, Dordrecht (1992);
``Comment on Dynamical Randomness in Quantum Systems",
Prog.\ Theor.\ Phys.\ Suppl.\ {\bf 116}, 369-378 (1994).

\bibitem{Kryl}
N.\ N.\ Krylov: {\it Works on the Foundations of Statistical Mechanics},
Princeton University Press (1979).

\bibitem{Jona}
%G.\ Jona-Lasinio, C.\ Presilla, and F.\ Cappaso:
%``Chaotic Quantum Phenomena without Classical Counterpart",
%Phys.\ Rev.\ Lett.\ {\bf 68}, 2269 (1992);
G.\ Jona-Lasinio and C.\ Presilla:
``Chaotic Properties of Quantum Many-Body Systems in the Thermodynamic Limit",
Phys.\ Rev.\ Lett.\ {\bf 77}, 4322 (1996).

\bibitem{JS}
R.\ V.\ Jensen and R.\ Shankar:
``Statistical Behavior in Deterministic Quantum Systems 
with Few Degrees of Freedom''
Phys.\ Rev.\ Lett.\ {\bf 54}, 1879 (1985);
K.\ Saito, S.\ Takesue, and S.\ Miyashita: 
J.\ Phys.\ Soc.\ Jpn.\ {\bf 65}, 1243- (1996).

\bibitem{YI}
H.\ Yamada and K.\ S.\ Ikeda: 
``Spontaneously induced irreversible energy transport
in closed electron-phonon systems'', Phys.\ Lett.\ A {\bf 222}, 76 (1996).

\bibitem{KSE}
G.\ Kopidakis, C.\ M.\ Soukoulis, and E.\ N.\ Economou:
``Localization and electron-phonon interactions in disordered systems",
Europhys.\ Lett.\ {\bf 33}, 459-464 (1996).

\bibitem{YIG}
H.\ Yamada, K.\ Ikeda and M.\ Goda: 
``Quantum diffusion in a coherently time-varying one-dimensional
disordered system'',
Phys.\ Lett.\ A {\bf 182}, 77 (1993).

%\bibitem{DH}
%K.\ Drese and M.\ Holthaus:
%``Exploring a Metal-Insulator Transition with Ultracold Atoms in
%Standing Light Waves ?",
%Phys.\ Rev.\ Lett.\ {\bf 78}, 2932 (1997).

\bibitem{Bran}
T.\ Brandes: unpublished.

%\bibitem{Lloy}
%P.\ Lloyd:
%J.\ Phys.\ C {\bf 5}, 1717- (1969).

\bibitem{Bayf}
J.\ E.\ Bayfield and P.\ M.\ Koch: 
``Multiphoton Ionization of Highly Excited Atoms",
Phys.\ Rev.\ Lett.\ {\bf 33}, 258 (1974).
%J.\ E.\ Bayfield, L.\ D.\ Garner, and P.\ M.\ Koch: ibid.\ {\bf 39}, 76 (1977).
%for a review of works of Bayfield's group, see
%J.\ E.\ Bayfield: ``Studies of the Sinusoidally Driven Weakly Bound Atomic Electron
%in the Threshold Region for Classically Stochastic Behavior",
%in {\it Quantum Measurement and Chaos}, ed. by R.\ E.\ Pike and S.\ Sarkar,
%Plenum, New York (1987).

\bibitem{CCSG}
%G.\ Casati, B.\ V.\ Chirikov, and D.\ L.\ Shepelyansky: 
%``Quantum Limitations for Chaotic Excitation of 
%the Hydrogen Atom in a Monochromatic Field'',
%Phys.\ Rev.\ Lett.\ {\bf 53}, 2525 (1984);
G.\ Casati, B.\ V.\ Chirikov, I.\ Guarneri, and D.\ L.\ Shepelyansky: 
``Relevance of Classical Chaos in Quantum Mechanics:
The Hydrogen Atom in a Monochromatic Field'',
Phys.\ Rep.\ {\bf 154}, 77-123 (1987).

\bibitem{Blum}
R.\ Bl\"umel, R.\ Graham, L.\ Sirko, 
U.\ Smilansky, H.\ Walther, and K.\ Yamada:
``Microwave Excitation of Rydberg Atoms in the Presence of Noise",
Phys.\ Rev.\ Lett.\ {\bf 62}, 341 (1989).

\bibitem{OA}
E.\ Ott, T.\ M.\ Antonsen, Jr, and J.\ D.\ Hanson:
``Effect of Noise on Time-Dependent Quantum Chaos",
Phys.\ Rev.\ Lett.\ {\bf 53}, 2187 (1984);
S.\ Adachi, M.\ Toda, and K.\ Ikeda:
``Potential for Mixing in Quantum Chaos'',
Phys.\ Rev.\ Lett.\ {\bf 61}, 655 (1988).

\bibitem{Thou}
D.\ J.\ Thouless:
Phys.\ Rev.\ Lett.\ {\bf 39}, 1167 (1977);
Solid State Commun.\ {\bf 34}, 683 (1980).

\bibitem{Moor}
F.\ L.\ Moore, J.\ C.\ Robinson, C.\ F.\ Bharucha, B.\ Sundaram, and M.\ G.\ Raizen: 
``Atom Optics Realization of the Quantum $\delta$-Kicked Rotor",
Phys.\ Rev.\ Lett.\ {\bf 75}, 4598 (1995).

%\bibitem{BEC}
%A.\ Griffin, D.\ W.\ Snoke, and S.\ Stringari (eds.):
%{\it BOSE-EINSTEIN CONDENSATION},
%Cambridge University Press, Cambridge (1995).


 
\end{thebibliography}

\maketitle
\section{Introduction}
A quantum kicked rotor (QKR) is one of the origins of {\it quantum chaos},
which roughly implies the field of study on {\it classically chaotic} quantum systems.
Since the first numerical investigation of a QKR by Casati {\it et al}. \cite{Casa},
there have appeared many descendants of it.
In order to settle the situation, I here review 
QKRs and the related systems which include a classical kicked rotor, 
an interacting boson system, an electron-phonon system and others.

\section{A classical kicked rotor (CKR)}

First of all, 
I discuss the properties of a {\it classical} kicked rotor (CKR).
The Hamiltonian for a CKR is,
\begin{eqnarray}
\label{HamCKR}
\mathfbox{
H(t) = p^2/2I +k V(\theta) \sum_n \delta(t-n),
}
\end{eqnarray}
where $p$ denotes an angular momentum, $I$ a moment of inertia, 
$K$ the strength of the nonlinearity,
and the period of the kick is normalized to be one.  
Note that this is a (periodically) {\it time-dependent} Hamiltonian,
and there is no {\it stochastic (noise) term} in it.
The equations of motion for a CKR {\it when $V(\theta)=\cos \theta$}, 
after some normalizations of the variables, are
\begin{eqnarray}
%\mathfbox{
\theta_{n+1} &=& (\theta_n +p_n) \hspace{2mm} {\rm mod} \hspace{1mm} 2 \pi,
\\
p_{n+1} &=& p_n+k \sin \theta_{n+1},
%}
\end{eqnarray}
where $n$ represents the discretized time step.
This is called {\it the standard map} \cite{Zasl,Ott}. 
Due to the delta function in Eq.\ (\ref{HamCKR}),
it {\it exactly} becomes a discrete map. 
So the numerical calculation is efficiently done.
Iterating the map, we can calculate, for example, the dynamics of the 
variance of $p$, and it was found that, under some conditions,
\begin{eqnarray}
\mathfbox{
\langle (p_n-p_0)^2 \rangle \simeq D n,
}
\end{eqnarray}
where the brackets mean the average for the initial ensemble,
and $D \simeq k^2/2$, which is numerically known.
This behavior is very similar to that of {\it diffusion}
[cf. $\langle (x_t-x_0)^2 \rangle \simeq D' t$], so 
this is called ``the diffusion of momentum".
This issue, which is understood as an example of the collapse of KAM tori and
the {\it spontaneous} emergence of stochasticity,
was intensively studied by Chirikov and his coworkers \cite{Chir}.

\section{A quantum kicked rotor (QKR)}

It is natural to ask what happens when one quantizes the CKR.
This was first numerically investigated by Casati {\it et al}. \cite{Casa}.
They employed the following 
Schr\"odinger equation for a {\it quantum} kicked rotor (QKR),
\begin{equation}
\label{QKR}
\mathfbox{
i \hbar \frac{\partial}{\partial t} \Psi(\theta,t)
= H(t) \Psi(\theta,t)
=\left[- \frac{\hbar^2}{2 I} \frac{\partial^2}{\partial \theta^2}
+k V(\theta) \sum_n \delta(t-n) \right] \Psi(\theta,t),
%= \{ H_0+\hat k V(\theta) \Delta(t)\} \Psi(\theta,t),
}
\end{equation}
where $H(t)$ is given by Eq.\ (\ref{HamCKR}) 
with the replacement $p=(\hbar/i) \partial/\partial \theta$.
Then, they numerically calculated the wavefunction of a QKR from 
Eq.\ (\ref{QKR}) as follows:
\begin{itemize}
\item
First, take the initial wave function. [e.g., $\Psi(\theta,t=0)=(2 \pi)^{-1/2}$.]
\item
Second, multiply the wave function by $\exp [-i(k/\hbar) V(\theta)]$.
\item
Fourier transform it [$\Psi(\theta) \rightarrow \Psi_l$].
\item
Multiply the Fourier transformed wave function by $\exp [-i \hbar l^2/2I]$.
\item
Inverse Fourier transform it [$\Psi_l \rightarrow \Psi(\theta)$].
\item
Return to the second step.
\end{itemize}
Iterating this loop, we obtain the dynamics of the wavefunction $\Psi(\theta,t)$,
and can calculate the quantum average of any physical quantity.
To discuss the relation between a CKR and a QKR, 
they calculated the variance of $p$ as follows,
\begin{equation}
\langle p_t^2 \rangle
=\hbar^2 \sum_{l=-\infty}^{\infty} l^2 |\Psi_l(t)|^2
\end{equation}
where the brackets mean the quantum average, and we have assumed, 
for simplicity, $\langle p_0 \rangle =0$, i.e., $\Psi(\theta,0)=(2 \pi)^{-1/2}$.
{\it When} $V(\theta)=\cos \theta$, they found, contrary to the case of a CKR, that
a QKR stops diffusing
\footnote{
As Izraelev and Shepelyansky showed \cite{IS},
for special values of $\hbar$,
there is the cases where $\langle p_t^2 \rangle \propto t^2$. 
This is called ``quantum resonance".}
in momentum space at a certain time $t^*$.
This result, {\it suppression of chaos in a quantum system}, seems to lead,
in general, to the absence of the {\it quantum stochasticity} \cite{Zasl,QS},
which denotes such concepts as mixing property  
or approaches to equilibrium states in quantum systems.
However, there are some ways to recover 
the stochasticity in quantum systems as discussed later.

\section{Mapping a QKR onto a tight-binding model}

Mapping is a useful concept when we understand something deeper.
In this section, following Fishman {\it et al}. \cite{Fish}, we map {\it one} QKR
onto another problem, which is related to 1D Anderson Localization (AL) \cite{Ishi}.
Since the issue on AL has been well studied in solid-state physics \cite{Lee},
we can, to understand QKRs, use the method developed in the AL literature,
and vice versa.

In the following, I review the procedure in detail:
The Hamiltonian in Eq.\ (\ref{QKR}) has the periodicity, $H(t+1)=H(t)$,
so the Floquet's theorem can be applied, i.e., the wavefunction can be written as 
$\Psi(\theta,t)=e^{-i \omega t}u_\omega (\theta,t)$, where
$u_\omega(\theta,t+1)=u_\omega(\theta,t)$ 
and $\omega$ is called a {\it quasi-energy} (which is like a Bloch index
in the field of solid-state physics \cite{Kitt}).
Hence, all we need is the dynamics from
$t=0_+$ (just after the kick at $t=0$) to $t=1$.
Since the particle freely moves between $t=0_+$ and $t=1_-$
(just before the kick at $t=1$),
the transformation is 
\begin{equation}
\label{free}
u_{-,m}=u_{+,m}e^{i E_m},
\end{equation}
where
$u_+(\theta)=u_+(\theta, t=0_+)$, $u_-(\theta)=u_-(\theta, t=1_-)$,
$E_m=\omega -\tau m^2/2$, $\tau=1/I$, and 
we have defined the Fourier component of $f(\theta)$, $f_m$, as:
\[
f(\theta) = \sum_m f_m e^{i m \theta}, \hspace{5mm}
f_m = \frac{1}{2\pi}\int_0^{2\pi} d\theta f(\theta) e^{-i m \theta}.
\]
Inverse Fourier transforming Eq.\ (\ref{free}), we obtain
\begin{equation}
\label{free2}
u_-(\theta)=\sum_m \int \frac{d \phi}{2 \pi} e^{i m(\theta-\phi)+i E_m} u_+(\phi).
\end{equation}
Between $t=1_-$ and $t=1_+=0_+$, 
the particle feels a kick and is transformed to be 
\begin{equation}
\label{kick}
u_+(\theta)=u_-(\theta) e^{-i k V(\theta)}.
\end{equation}
Next, we define $\bar u = (u_+ +u_-)/2$, and have, from Eq.\ (\ref{kick}),
$\bar u = u_+/(1-i U)=u_-/(1+i U)$, where $U=-\tan[k V(\theta)/2]$.
Substituting these into Eq.\ (\ref{free2}) to eliminate $u_+, u_-$,
we have
\begin{eqnarray}
[1-i U(\theta)] \bar u(\theta)
&=&
\sum_m \int \frac{d \phi}{2\pi} e^{i m(\theta-\phi)+i E_m}
[1+i U(\phi)] \bar u(\phi)
\nonumber
\\
&=&
\sum_m e^{i (E_m +m \theta)} \int \frac{d \phi}{2\pi}
\left[ 
\sum_r u_r e^{i(r-m) \phi}+i \sum_n \sum_r U_n u_r e^{i(n+r-m) \phi} 
\right]
\nonumber
\\
&=&
\sum_m e^{i (E_m +m \theta)}
\left[
u_m +i \sum_r U_{m-r} u_r 
\right].
\nonumber
\end{eqnarray}
Fourier transforming the above equation, 
we obtain the left-hand side (l.h.s.) and right-hand side (r.h.s.) of the equation as
\begin{eqnarray}
{\rm l.h.s.}
&=&
u_n -i \sum_r U_{n-r} u_r,
\nonumber
\\
{\rm r.h.s.}
&=&
e^{iE_n} (u_n +i \sum_r U_{n-r} u_r).
\nonumber
\end{eqnarray}
Equating these, we finally have
\begin{equation}
\label{AL}
%\tan(E_n/2) u_n +\sum_r U_{n-r} u_r =
\mathfbox{
T_n u_n +\sum_r U_{n-r} u_r =0,
}
\end{equation}
where $T_n =\tan(E_n/2)$.
This is just the tight-binding model in solid-state physics \cite{Kitt},
if we regard the subscript ``r" of $u_r$ as 
a ``site" in {\it momentum} space,
$T_n$ as a site energy, and $U_r$ as a hopping energy.
In the AL literature, the site energy $T_n$ is a stochastic variable,
whereas it is just a {\it deterministic function} of $n$ here.
In spite of this fact, if $\tau$ is an {\it irrational number},
$T_n =\tan(E_n/2)$ behaves {\it like random},
\footnote{
This is also called {\it quasi-random} (or {\it pseudo-random}).
In fact, since, in numerical calculations, 
some algorithms must be used to generate random numbers,
{\it only} quasi-random numbers are available for us.
As Kolmogorov, Chaitin, and Solomonoff independently showed,
we, {\it in  principle}, can {\it not} judge which number is {\it exactly} random
though we {\it can} define randomness \cite{Chai}.
This fact is,  as the halting problem of a (universal) Turing machine \cite{Penr},
another version of G\"odel's incompleteness theorem \cite{Gell}.}
and AL {\it does} appear even if Eq.\ (\ref{AL}) is employed!
This is a very interesting and striking result 
that Fishman {\it et al}. showed in their numerical calculations.

On the other hand, if $\tau$ is a rational number,
$T_n$ becomes a periodic function of $n$.
Hence, all the states become extended (Bloch) ones, 
in other words, AL disappears.
They considered that this case is rather rare in physics, 
and that AL always will appear.
However, because they did not find an {\it exact} link between a QKR and 1D AL,
their argument, in a sense, was not a complete one.
Recently, Altland and Zirnbauer \cite{AZ} showed that the behaviors of a QKR 
{\it in low-frequency regions} can be mapped onto 
a supersymmetric nonlinear sigma (SUSYNS) model.
On the other hand, it is known that AL problem {\it without
electron-electron interactions} also can be mapped onto the same model \cite{Efet}.
Therefore, the relation, 1D AL $\sim$ a SUSYNS model $\sim$ a QKR was established.
Strictly speaking, however, their argument only concerned with 
a link between {\it one} QKR and 1D AL. 
What happens when we extend one-body QKR to 
many-body (or higher dimensional) ones ?

\section{Extension to many-body QKR systems}

There are many ways to extend a QKR to many-body ones.
One possibility is the following Hamiltonian,
\begin{equation}
\label{HamMB}
\mathfbox{
H(t)=
- \frac{\hbar^2}{2 I} \sum_i \frac{\partial^2}{\partial \theta_i^2}
+k V(\{\theta_k \}) \sum_n \delta(t-n),
}
\end{equation}
where $V(\{\theta_k \})=V(\theta_1, \theta_2,..., \theta_N)$ can be complex.
In particular, the following form of the potential energy,
\begin{eqnarray}
\mathfbox{
V(\{\theta_k \}) 
%&=& \sum_i V^{(1)}(\theta_i) + \sum_{i \neq j} V^{(2)}(\theta_i - \theta_j)
%\nonumber
%\\
= \sum_{i=1}^N K_i \cos \theta_i +
\sum_{i \neq j}^N \epsilon_{ij} \cos (\theta_i-\theta_j),
}
\end{eqnarray}
was examined by Toda, Adachi, and Ikeda \cite{TAI}. 
In a physical sense, this is a natural choice because it resembles
a Heisenberg XY model in the field of magnetism.
They showed in their numerical calculations that
even if $N$ (the total number of QKRs) $=2$, there appears, in a sense, 
again the diffusion of the momentum which exists in a CKR. 

Now let us map this ($N=2$-QKR) problem onto 2D AL.
This is a generalization of the method by Fishman {\it et al.}:
\begin{equation}
%\label{AL}
%\tan(E_n/2) u_n +\sum_r U_{n-r} u_r =
T_{\bf n} u_{\bf n} 
+\sum_{\bf r} U_{\bf n-r} u_{\bf r} =0,
\end{equation}
where ${\bf n} =(n, m)$ is a {\it pair} of integers.
If we use the analogy with 2D AL, there should be 
localization due to the scaling argument \cite{Four}.
According to Toda {\it et al.}, however, {\it there is not !}
This remarkable discrepancy, I think, deserves further considerations.
{\bf We need to calculate the localization length 
and/or something else in 2D (or higher) momentum space},
like Shepelyansky did in the 1D space \cite{Shep}.
We also must investigate how the property of the hopping integral
$U_{\bf n-r}$ affects the result. In the textbook, it is often assumed
that $U_{\bf n-r}$ has only nearest-neighbor components,
but it is not the case in general.
\footnote{
Of course, after renormalization procedure, there might appear a universality
where {\it only} nearest-neighbor hopping is relevant.}

Furthermore, it is known that a 3D-random system exhibits 
the metal-insulator (MI) transition. 
So, {\bf what is the correspondence of the MI transition
in quantum chaos ?} %\cite{Bran}
This issue was considered by Casati {\it et al}. \cite{CGS}
in the following many-body QKR model,
\begin{equation}
\label{Casa}
H(t)=H_0(n)+\omega_1 n_1 +\omega_2 n_2 
+V(\theta, \theta_1, \theta_2) \sum_n \delta(t-n),
\end{equation}
where $H_0(n)$ is a random Hamiltonian (this might be a trick !),
$n_i=-i\partial/\partial \theta_i$ ($i$=1,2), and $\omega_1, \omega_2$ 
incommensurate frequencies. 
There are three degrees of freedom, $\theta, \theta_1, \theta_2$, 
so we can map this onto 3D AL. They, in particular, took the following potential,
\[
V(\theta, \theta_1, \theta_2)
=-2 \tan^{-1} [2 k(\cos \theta+\cos \theta_1+\cos \theta_2)],
\]
for the mapped model to have {\it only} nearest-neighbor hopping.
%\[
%T_{\bf n} u_{\bf n} +k {\sum_{\bf n}}' u_{\bf n}=0,
%\]
%where $T_{\bf n}=-\tan[\frac{1}{2}(E_n+n_1 \omega_1 +n_2 \omega_2+\lambda)]$
%($\lambda$ is quasienergy, $E_n$ random energy distributed in $[0, 2 \pi]$)
%and the sum $\sum'$ includes only nearest-neighbor hopping,
%is easier to calculate. 
They then numerically calculated the localization length in the case of localization,
and the diffusion of momentum in the case of delocalization, and
showed that there occurs the localization-delocalization transition corresponding to 
the MI transition in 3D-random systems. 

There exist other generalizations of a QKR \cite{Shep2,Iked}.
Since the number of the generalizations is, of course, enormous,
{\bf we need some phenomenology to classify many-body QKR systems} 
for economic reasons. This issue remains to be investigated.

\section{Many-body quantum systems except QKRs}

Next, we will consider some many-body systems except QKRs.
The first example is the {\it quantum hard sphere gas} \cite{Gasp}, 
which is described by the following Schr\"odinger equation,
\begin{equation}
-\frac{\hbar^2}{2 m} \sum_a \nabla_a^2 \Psi =E \Psi,
\end{equation}
where $\Psi=0$ at the boundary defined by
%\begin{eqnarray}
\[
| {\bf r}_a -{\bf r}_b | \geq d,
\hspace{2mm}
d/2 \leq x_a, y_a, z_a \leq L-d/2.
\]
%\end{eqnarray}
Its classical counterpart was studied by Krylov and Sinai \cite{Kryl},
who showed that the classical hard sphere gas has positive 
{\it Kolmogorov-Sinai entropy}, 
which is one of the indicators of stochasticity in classical physics. 
For the semiclassical treatment of quantum hard sphere gas,
see Ref. \cite{Gasp}. 

The second is the interacting boson system studied by 
Jona-Lasinio and Presilla \cite{Jona}.
Their Hamiltonian is 
\begin{equation}
\label{Jona}
H=\sum_{j=1}^L [\alpha_j a_j^{\dagger} a_j
-\beta_j(e^{i \theta} a_{j+1}^{\dagger} a_j +{\rm h.c.})]
+\sum_{j=1}^L \gamma_j a_j^{\dagger} a_j^{\dagger}a_j a_j,
\end{equation}
where the operator $a^{\dagger}_j$ creates a boson in the site $j$,
and $\alpha_j, \beta_j$ and $\gamma_j$ are the site, hopping, and interaction 
energies, respectively.
They numerically calculated the following quantity,
\begin{equation}
\kappa_{AB}=\lim_{t \rightarrow \infty}
\sqrt{\frac{1}{t}\int_0^t dt'
|\langle A(t) B \rangle - \langle A \rangle \langle B \rangle|},
\end{equation}
which indicates the potential of a mixing property, i.e., the maximal mixing implies
that $\kappa_{AB}=0$.
They showed that as the number of the particles increases 
(although the calculated number is only up to seven),
$\kappa_{AB}$ decreases to zero. This result agrees with our intuition.
They also studied its classical counterpart with mean-field approximation.
The approximated equation of motion derived from Eq.\ (\ref{Jona}) is
\begin{equation}
\label{Jona2}
i \hbar \frac{d}{dt} z_j(t)
=[\alpha_j +2(N-1) \gamma_j |z_j(t)|^2] z_j(t)
-\beta_{j-1} e^{i \theta} z_{j-1}(t) 
-\beta_{j} e^{-i \theta} z_{j+1}(t),
\end{equation}
where $z_j(t)$ is the amplitude of the mean field in the site $j$.
They numerically showed that when the classical system described 
by Eq.\ (\ref{Jona2}) becomes chaotic {\it in a classical sense},
the corresponding quantum system described by Eq.\ (\ref{Jona}) has a mixing property.
This is one of the examples of {\it quantum-classical correspondence} 
studied in the field of quantum chaos.
Similar {\it spin} systems were also treated by Jensen, Shankar
and others \cite{JS}, and it was shown that even {\it closed small}
integrable (and, of course, non-integrable) systems {\it numerically} exhibit 
a mixing property.

The third is the 1D-random electron system coupled with {\it a few modes} 
of phonon,
which was studied by Yamada and Ikeda \cite{YI}. Their Hamiltonian is 
\begin{eqnarray}
\label{YI}
H &=&H_{\rm el}+H_{\rm ph}+H_{\rm int},
\nonumber
\\
&=&\sum_{n=1}^N |n \rangle V(n) \langle n | 
+ \sum_{n \neq m}^N |n \rangle K(n,m) \langle m |
\nonumber
\\
&&+\sum_{j=1}^M (\frac{1}{2} \hat p_j^2 +\frac{1}{2} \omega_j^2 \hat q_j^2)
+\sum_{n=1}^N \sum_{j=1}^M  [|n \rangle V(n) \langle n |]b_j \hat q_j,
\end{eqnarray}
where $V(n)$ is the random site energy distributed in $[-W,W]$,
 $K(n,m)$ the hopping energy,
$\hat q_j, \hat p_j$ the coordinate and momentum of phonons, respectively,
$b_j$ the coupling strength between the electron and phonons.
(A similar, but less quantum, model was examined 
by Kopidakis {\it et al}. \cite{KSE}. 
One may point out the similarity between Eq.\ (\ref{Casa}) and above.)
They numerically calculated from Eq.\ (\ref{YI}) 
the mean-square displacement, 
\[
\langle \Delta \hat n^2 \rangle 
= \langle \Psi(t)| (\hat n - \langle \hat n \rangle)^2 | \Psi(t) \rangle,
\]
where $\hat n=\sum_{j=1}^N j |j \rangle \langle j|$ is the position operator
and $| \Psi(t) \rangle$ is the time-dependent wave function.
This quantity characterizes whether the electron becomes localized or extended.
Their numerical experiments showed that there is an anomalous diffusion law,
$\langle \Delta \hat n^2 \rangle =D t^{\alpha}$,
and $\alpha$ approaches one from below 
as the number of the phonon mode increases.
(When the number is {\it three}, we can conceive normal diffusion:
$\langle \Delta \hat n^2 \rangle \simeq D t$.)
At the same time, they calculated the energy of the electron system,
$E_{\rm el}(t)=\langle \Psi(t)| H_{\rm el}| \Psi(t) \rangle$,
and that of the phonon system, 
$E_{\rm ph}(t)=\langle \Psi(t)| H_{\rm ph}| \Psi(t) \rangle$.
They showed, when the coupling $b_j$ 
and the randomness $W$ are strong enough, there exists 
``a one-way quasi-stationary energy flow from electron to phonons" \cite{YI}.

Next, we see, in Yamada-Ikeda model, what happens when all the phonon modes 
are excited in coherent states, i.e., all the phonons are {\it classicalized}.
\footnote{
They instead considered the case when all the phonon modes 
are excited around {\it Fock state} with a large quantum number \cite{YI}.
But, this, I think, is a rather unphysical choice, because 
the states are highly nonclassical.}
%Then, using the same procedure 
%in the field of quantum optics \cite{Loud}, we can map Eq.\ (\ref{YI})
%onto the time-dependent Hamiltonian which does not contain 
%the degrees of freedom of the phonons.
The procedure is as follows:
Unitary transform the Schr\"odinger equation 
$i \hbar \frac{\partial}{\partial t} \Psi(t)
= (H_{\rm el}+H_{\rm ph}+H_{\rm int}) \Psi(t)$
into another one for $\phi(t)=\exp(i H_{\rm ph} t/\hbar) \Psi(t)$,
which satisfies,
\[
i \hbar \frac{\partial}{\partial t} \phi(t)
= [H_{\rm el}+ 
e^{i H_{\rm ph} t/\hbar} H_{\rm int} e^{-i H_{\rm ph} t/\hbar}]\phi(t).
\]
We can expand $e^{i H_{\rm ph} t/\hbar} H_{\rm int} e^{-i H_{\rm ph} t/\hbar}$ as
\begin{eqnarray}
e^{i H_{\rm ph} t/\hbar} H_{\rm int}e^{-i H_{\rm ph} t/\hbar}
&=&\sum_{n=1}^N \sum_{j=1}^M  [|n \rangle V(n) \langle n |]b_j 
\sqrt{\frac{\hbar}{2 \omega_j}}(\hat a_j e^{-i \omega_j t} +{\rm h.c.})
\nonumber
\\
&\simeq& 
\sum_{n=1}^N \sum_{j=1}^M  [|n \rangle V(n) \langle n |]
\epsilon_j \cos(\omega_j t+\phi_j),
\end{eqnarray}
where
\begin{eqnarray}
\epsilon_j &=&
\sqrt{2 \hbar b_j^2 |\alpha_j|^2/\omega_j},
\hspace{2mm}
\hat q_j = \sqrt{\hbar/2 \omega_j}(\hat a_j+\hat a_j^{\dagger}),
\nonumber
\\
H_{\rm ph}
&=& \sum_{j=1}^M \hbar \omega_j (\hat a_j^{\dagger}\hat a_j+1/2),
\hspace{2mm}
e^{i H_{\rm ph} t/\hbar} \hat a_j e^{-i H_{\rm ph} t/\hbar}
= \hat a_j e^{-i \omega_j t},
\nonumber
\end{eqnarray}
and we have used the following {\it approximation},
\[
\hat a_j \simeq  |\alpha_j| e^{-i \phi_j},
\]
(where $\alpha_j, \phi_j$ are c-numbers)
which is justified when all the phonons are {\it classicalized}
due to decoherence or something.
Hence, the reduced system is described by
\[
H'_{\rm el}
=\sum_{n=1}^N |n \rangle V(n,t) \langle n | 
+ \sum_{n \neq m}^N |n \rangle K(n,m) \langle m |,
\]
where
\[
V(n,t)=V(n)[1+\sum_{j=1}^M \epsilon_j \cos(\omega_j t+\phi_j)].
\]
This system, with incommensurate frequencies $\omega_j$, 
was also investigated by them and Goda \cite{YIG}.
It was shown that the periodic perturbation causes the destruction of 
localization.
%which is similar to the behaviors of Harper's model,
%\[
%H_{\rm Harper}
%=\sum_{n=1}^N |n \rangle \nu(n) \langle n | 
%+ \sum_{n \neq m}^N |n \rangle K(n,m) \langle m |,
%\]
%where 
%$\nu(n)=\nu_0 \cos(2 \pi g n+\eta)$, $g$ is an irrational number, and 
%$\eta$ a phase. (An atom-optics realization of this model was discussed 
%in Ref.\ \cite{DH}.) 
A little different system in {\it more than two dimensions}
is being examined by Brandes \cite{Bran}.

The systems discussed above are only a few examples of 
many-body quantum chaos systems.
There will be, of course, other interesting systems to study 
the quantum stochasticity.


\section{Experimental aspects of quantum chaos systems}

In this section, I discuss the experimental aspects of quantum chaos systems.
%which is, e.g., observed in a QKR up to a certain time $t^*$ as discussed above.
From a historical point of view, 
the results of experiments \cite{Bayf} on ionization of Rydberg atoms by an
external field were first shown to have the properties of a quantum chaos
system \cite{CCSG}.
The Hamiltonian considered is
\begin{equation}
H_0(t)=\frac{p^2}{2m}-\frac{e^2}{4 \pi \epsilon_0} \frac{1}{x}
+e x F \sin(\omega t), \hspace{2mm} x>0,
\end{equation}
where $F$ and $\omega$ are the field strength and frequency in atomic units.
This is a model rather different from a QKR, but it has similar consequences,
including quantum limitation of chaotic excitation (ionization). 
In real situations, of course, {\it noise} affects the behaviors of the system considered.
This issue was investigated by Bl\"umel {\it et al}. \cite{Blum}.
Their total Hamiltonian, including the variables for the environment, is 
\[
H(t)=H_0(t)+H_{\rm int}+H_{\rm R},
\]
with 
%\begin{eqnarray}
\[
H_{\rm int}=\hbar \sum_i g(\omega_i) x (b_i +b_i^{\dagger}),
\hspace{2mm}
H_{\rm R}=\hbar \sum_i \omega_i (b_i^{\dagger}b_i+1/2).
\]
%\end{eqnarray}
Their central result is that small amount of noise causes the destruction of 
quantum coherence and the localization in energy space, which is the origin of 
quantum limitation of chaotic excitation, disappears.
This is very similar to the results in the case of a QKR \cite{OA}.
In solid-state physics literature, 
this is similar to those by Thouless \cite{Thou} who considered the effect of 
finite temperature on localization in a solid.

Those discussed above are not directly related to QKRs.
But, in the field of atom optics, there exists an ideal realization of a QKR \cite{Moor}.
The situation is as follows:
An atom is exposed to the delta-pulse trains of light,
which simultaneously simulate the potential and the delta functions in Eq.\ (\ref{QKR}). 
It was shown that the atom-optics system of a QKR
{\it experimentally} exhibits 
the same behavior as predicted by Casati {\it et al}. \cite{Casa}.

Atom-optics systems are now being studied 
in connection with Bose-Einstein condensation. %\cite{BEC}.
I think {\bf these systems are also promising, 
both experimentally and theoretically, 
to investigate quantum chaos sytems, including many-body QKR ones, 
and to do quantum stochasticity}.


\section{Summary}
Quantum kicked rotors (QKRs) are typical examples in the field of quantum chaos.
A QKR causes the suppression of chaos in very short time scale,
and that is all. However, if there are more degrees of freedom, many things can happen.
We need more calculations for many-body QKR systems,
and some phenomenology to classify them.
In order to achieve these, it is important 
to map them onto other systems (including tight-binding models,
a supersymmetric nonlinear sigma model, and many others),
because the crossreference which occurs there makes our understanding deeper.
Atom-optics systems seem to be experimentally best
to test the theoretical calculations on QKRs.
Of course, except QKRs, there are many other quantum chaos systems,
and many things remain to be done.



\end{document}
